Author’s response to reviews

Title: Intraductal papillary mucinous neoplasms of the pancreas and European guidelines: importance of the surgery type in the decision-making process

Authors:
Etienne Buscail (ebuscail@me.com)
Thomas Cauvin (cauvin.lapeyre@gmail.com)
Benjamin Fernandez (fernandezbenjamin@icloud.com)
Camille Buscail (camille.debrauer@gmail.com)
Marion Marty (marion.marty@chu-bordeaux.fr)
Bruno Lapuyade (bruno.lapuyade@chu-bordeaux.fr)
Clément Subtil (clement.subtil@chu-bordeaux.fr)
Jean-Philippe Adam (jean-philippe.adam@chu-bordeaux.fr)
Véronique Vendrely (veronique.vendrely@chu-bordeaux.fr)
Sandrine Dabernat (sandrine.dabernat@u-bordeaux.fr)
Christophe Laurent (christophe.laurent@chu-bordeaux.fr)
Laurence Chiche (laurence.chiche@chu-bordeaux.fr)

Version: 2 Date: 08 Aug 2019

Author’s response to reviews:

Editor section visceral and general surgery of BMC surgery

Bordeaux, 8 August, 2019
Dear Professor René Fahrner, dear Pr Kostiantyn Kopchak, dear Pr Megan Barling,

We are now submitting a revised version of our manuscript number BSUR-D-19-00376.

We have revised the manuscript according to editors comments.

We have followed carefully your instructions for the authors and hope that our revised manuscript will meet your expectations.

Best regards

Dr Etienne Buscail, MD, PhD